% !TEX program = pdflatex
% !TEX spellcheck = en_GB
% !TEX outputDirectory = tex_output
% !TEX enableSynctex = true
% !TEX jobNames = presentation

\makeatletter
\def\input@path{{/home/user_name/MyLatexTemplates/}{../}}
\newcommand{\maintitle}{A Nice Presentation Title}
\newcommand{\mainsubject}{Example Presentation}
\newcommand{\mainkeywords}{Keywords}
\makeatother

% Uncomment if Place Table of Contents only at start
% By default TOC outline will be placed at every section 
%\newcommand{\nosectiontoc}{}

% Uncomment to disable hyperref package
%\newcommand{\nolinks}{}

% Please un comment one of the two lines for 4:3 or 16:9 paper size
%\documentclass[10pt,aspectratio=43]{beamer}
\documentclass[10pt,aspectratio=169]{beamer}

\input{beamer.preamble.cls}

\begin{document}
\bstctlcite{IEEEexample:BSTcontrol}

%% Put the presentation title  here
\title{\maintitle}
\disablefooter
%% Put Author and Institute details here
\author{\FirstName~\LastName}
\institute{\Organisation \\
\href{\Website}{\Website}}%

\maketitle
\inserttoc

\enablefooter{\maintitle}

%% Slides Start Here each slide should be made as a frame
\section{Example Section1}

\begin{frame}{An exemplary slide}
"If you want to accomplish something in the world, idealism is not enough - you need to choose a method that works to achieve the goal." - Stallman \cite{refernce1,refernce2,refernce3,refernce4}
\end{frame}

\begin{frame}{An Item List with a Figure}
\itemcolumn{%
\item item 1
\item item 2
\item item 3
}{%
\begin{figure}
\includegraphics[width=0.32\paperwidth]{Style/phone.pdf}
\caption{A nice telephone}
\end{figure}
}
\end{frame}

\section{Example Section2}
\begin{frame}
\end{frame}

\begin{frame}[allowframebreaks]
\frametitle{References}
\footnotesize
\bibliographystyle{IEEEtran}%
\bibliography{IEEEabrv,IEEEbst,local}%
\end{frame}

\end{document}
